Author’s response to reviews

Title: UK adults’ implicit and explicit attitudes towards obesity: a cross-sectional study

Authors:

Stuart W Flint (S.Flint@shu.ac.uk)
Joanne Hudson (Joanne.Hudson@leedsbeckett.ac.uk)
David Lavallee (david.lavallee@stir.ac.uk)

Version: 2
Date: 20 February 2015

Author’s response to reviews:

The four revisions as suggested have now been made. They were to:

1. Please include email address of all authors on the title page. All email addresses now appear on the title page.

2. Competing interests: Please include a 'Competing interests' section between the Conclusions and Authors’ contributions. This has been included.

3. Authors’ contributions: Please include an Authors’ contributions section before the Acknowledgements and Reference list. This has been included.

4. Acknowledgements: We strongly encourage you to include an Acknowledgements section between the Authors’ contributions section and Reference list. This has been included.